\begin{thebibliography}{10}

\bibitem{Zena93}
Z.~Ariola.
\newblock {Relating Graph and Term Rewriting via B\"ohm Models}.
\newblock In {\em Proc. RTA '93, Montreal, Canada, June}, 1993.

\bibitem{Zena89}
Z.~Ariola and Arvind.
\newblock {P-TAC: A Parallel Intermediate Language}.
\newblock In {\em Proc. ACM Conference on Functional Programming Languages and
  Computer Architecture, London, UK}, September 1989.

\bibitem{Zena91a}
Z.~Ariola and Arvind.
\newblock {A Syntactic Approach to Program Transformations}.
\newblock In {\em Proc. ACM SIGPLAN Symposium on Partial Evaluation and
  Semantics Based Program Manipulation, New Haven, CT}, June 1991.

\bibitem{Zena92b}
Z.~Ariola and Arvind.
\newblock {Compilation of Id}.
\newblock In {\em Proc. of the Fourth Workshop on Languages and Compilers for
  Parallel Computing, Santa Clara, CA, Springer-Verlag LNCS 589}, August 1991.

\bibitem{Zena93d}
Z.~M. Ariola and J.~W. Klop.
\newblock {Equational Term Graph Rewriting}.
\newblock Technical Report Draft, 1993.

\bibitem{Barendregt}
H.~Barendregt.
\newblock {\em {The Lambda Calculus: Its Syntax and Semantics}}.
\newblock North-Holland, Amsterdam, 1984.

\bibitem{Dutch86}
H.~Barendregt, T.~Brus, M.~van Eekelen, J.~Glauert, J.~Kennaway, M.~van Leer,
  M.~Plasmeijer, and M.~R. Sleep.
\newblock {Towards an Intermediate Language based on Graph Rewriting}.
\newblock In {\em Proceedings of the PARLE Conference, Eindhoven, The
  Netherlands, Springer-Verlag LNCS 259}, June 1987.

\bibitem{Barendregt87c}
H.~Barendregt, M.~van Eekelen, J.~Glauert, J.~Kennaway, M.~Plasmeijer, and
  M.~Sleep.
\newblock {Term Graph Rewriting}.
\newblock In {\em Proceedings of the PARLE Conference, Eindhoven, The
  Netherlands, Springer-Verlag LNCS 259}, June 1987.

\bibitem{Bru87}
T.~Brus, M.~van Eekelen, M.~vam Leer, and M.~Plasmeijer.
\newblock {Clean - A language for Functional Graph Rewriting}.
\newblock In {\em Proc. ACM Conference on Functional Programming Languages and
  Computer Architecture, Portland, OR, Springer-Verlag LNCS 274}, 1987.

\bibitem{Hennessy}
M.~Hennessy.
\newblock {\em {Algebraic Theory of Processes}}.
\newblock MIT Press, 1988.

\bibitem{Huet2}
G.~Huet and J.-J. L\'evy.
\newblock {Computations in Orthogonal Rewriting Systems II}.
\newblock Technical report, INRIA, Rocquencourt, France, 1990.

\bibitem{Hughes82}
R.~J.~M. Hughes.
\newblock {Super-combinators}.
\newblock In {\em Proceedings of Lisp and Functional Programming, Pittsburg,
  PA}, 1982.

\bibitem{Johnsson85}
T.~Johnsson.
\newblock {Lambda Lifting: Transforming Programs to Recursive Equations}.
\newblock In {\em Proceedings of Functional Programming Languages and Computer
  Architecture, Nancy, France, Springer-Verlag LNCS 201}, September 1985.

\bibitem{Vinod90}
V.~Kathail.
\newblock {\em {Optimal Interpreters for Lambda-calculus Based Funtional
  Languages}}.
\newblock PhD thesis, Dept. of Electrical Engineering and Computer Science,
  MIT, Cambridge, MA, May 1990.

\bibitem{Klop91}
J.~Kennaway, J.~Klop, M.~Sleep, and F.~de~Vries.
\newblock {Transfinite Reductions in Orthogonal Term Rewriting Systems}.
\newblock In {\em Proc. RTA '91, Springer-Verlag LNCS}, 1991.

\bibitem{Kennaway90}
R.~Kennaway.
\newblock {Implementing Term Rewrite Languages in Dactl}.
\newblock {\em Theoretical Computer Science}, Vol. 1, 1990.

\bibitem{Lamping90}
J.~Lamping.
\newblock An algorithm for optimal lambda calculus reduction.
\newblock In {\em Proc. ACM Conference on Principles of Programming Languages,
  San Francisco, CA}, January 1990.

\bibitem{Levy78b}
J.-J. L\'{e}vy.
\newblock {\em {R\'{e}ductions Correctes et Optimales dans le Lambda-Calcul}}.
\newblock PhD thesis, Universite Paris VII, Paris, France, October 1978.

\bibitem{Nikhil90c}
R.~S. Nikhil.
\newblock {Id (Version 90.0) Reference Manual}.
\newblock Technical Report CSG Memo 284-a, Laboratory for Computer Science,
  MIT, Cambridge, MA, USA, July 1990.

\bibitem{Wadsworth}
C.~Wadsworth.
\newblock {\em {Semantics And Pragmatics Of The Lambda-Calculus}}.
\newblock PhD thesis, University of Oxford, Oxford, UK, September 1971.

\bibitem{Welch75}
P.~Welch.
\newblock {Continuous Semantics and Inside-out Reductions}.
\newblock In {\em $\lambda$-Calculus and Computer Science Theory, Italy,
  Springer-Verlag LNCS 37}, March 1975.

\end{thebibliography}

\section{Introduction}

Sharing of subexpressions is of utmost importance in the
implementation of functional languages.  Consider the function
definition ${\sf F} \; x = x + x$ and the expression ${\sf F}(2+3)$.
Any decent implementation, independently of the evaluation strategy
(normal-order or applicative-order) it employs, will evaluate the
subexpression $2+3$ only once.  Several compiler optimizations are
about increasing the sharing of subexpressions to avoid their repeated
evaluation.  In this paper, we discuss the syntactic and semantic
properties of a calculus, which is adequate for capturing the sharing
of subexpressions in first-order functional languages.  The results of
this paper are also relevant to compiling higher-order functional
languages, because compilers of such languages often employ a
technique known as ``lambda-lifting" \cite{Johnsson85}.
The program that results after
lambda-lifting is in a ``supercombinatory" form, and is treated as a
first-order program \cite{Hughes82}.

A way to capture sharing is to represent the expression as a graph
instead of a linear text string or tree.  This allows sharing of
identical terms through pointers, and avoids repeated evaluation of
identical terms as it is commonly done in normal-order reduction.
Graph reduction for the $\lambda$-calculus was proposed by Wadsworth
in order to bring together the advantages of both the applicative and
the normal order evaluation \cite{Wadsworth}.  Wadsworth also formally
proved the correctness of his graph reduction technique. As an
aside, Wadsworth also showed that his graph reduction did not capture
enough sharing to lead to an optimal interpreter.  More recently a new
graph structure, which allows sharing of ``{\it contexts}'', has been
proposed in \cite{Vinod90,Lamping90}.  This latter technique leads to
provably optimal interpreters for the $\lambda$-calculus
\cite{Levy78b}.  In this paper, however, we are not concerned with
optimality questions, and we restrict our attention to
``argument sharing'' in a language which is simpler 
than the $\lambda$-calculus.

Much of the past work on graph rewriting has been to prove its
correctness with respect to either the $\lambda$-calculus
\cite{Wadsworth} or Term Rewriting Systems
\cite{Dutch86,Barendregt87c,Bru87,Kennaway90}.  In contrast, this
paper explores graph rewriting as a system in its own right, and makes
no attempt to prove the correctness of a graph implementation with
respect to a ``tree (or unshared) view" of the computation.  Motivated
by what we have observed in real implementations of functional
languages, we explore syntactic and semantic properties of graphs with
cycles, and rewriting rules that recognize or create cycles.  In this
respect our calculus goes farther than either \cite{Wadsworth} or
\cite{Barendregt87c} where only acyclic graphs are considered.
Without cyclic graphs some important implementation ideas are ruled
out.  More recently, Klop {\it et al}. \cite{Klop91} have extended the
Barendregt's graph rewriting system to deal with cycles. However,
their approach is significantly different from ours in that they model
cyclic graph rewriting as ``transfinite reduction'' of infinitary
graph terms.

In the following, we formally introduce a Graph Rewriting System
(GRS). The basic feature of a GRS is the {\it block} construct, \ie
{\it letrec}. A block in a GRS is not treated as syntactic sugar for
application; it is central to expressing the sharing of
subexpressions. Our GRS includes cyclic terms and permits both {\em
non left-linear rules} and {\em left-cyclic rules}.  We prove that in
the absence of {\em interfering rules} a GRS is confluent.  This is a
more general result than the confluence theorem in \cite{Klop91}.  We
think that our approach also leads to a simpler proof of confluence
than in \cite{Klop91}.

We also develop a term model for a GRS without interfering rules along
the lines of L\'evy's term model for the $\lambda$-calculus.  We
introduce the notion of ``information content" associated to a term,
and show that the information content defines a congruence on the set
of terms. This result can be applied in a straightforward manner to
show the partial correctness of those optimizations that simply
increase the sharing in a term. An example of such an optimization is
the common subexpression elimination.  Moreover, the result implies
the partial correctness of the cyclic implementation of the ${\sf
Y}$-rule.

The paper is organized as follows: In Section \ref{crs}, we introduce
graph rewriting taking combinatory logic as an example, and compare our
notation with that of Barendregt's \cite{Barendregt87c}.
We formally describe GRSs, in Section \ref{formalGRS}.
We introduce an
ordering on terms based on sharing of subterms, and formulate the
notion of a redex using this ordering.  In Section
\ref{orthogonal}, we introduce GRSs without interfering rules and
prove their confluence. Section \ref{equivalence} gives a term model
for the semantics of non-interfering GRSs.  In Section \ref{opti}, we
apply the results of Section \ref{equivalence} to prove the
correctness of some compiler optimizations, while in Section
\ref{conclusions} we discuss directions for future work.
